Author's response to reviews

Title: Study protocol. Active video games to promote physical activity in children with cancer: a randomized clinical trial with follow-up

Authors:

Lotta Kauhanen (anloka@utu.fi)
Liisa Järvelä (lisula@utu.fi)
Päivi M Lähteenmäki (paivi.maria.lahteenmaki@tyks.fi)
Mikko Arola (mikko.arola@pshp.fi)
Olli J Heinonen (olli.heinonen@utu.fi)
Anna Axelin (anmaax@utu.fi)
Johan Lilius (johan.lilius@abo.fi)
Tero Vahlberg (tervah@utu.fi)
Sanna Salanterä (sansala@utu.fi)

Version: 2 Date: 20 September 2013

Author's response to reviews:

Editor-in-Chief

20th September, 2013

Dear Executive Editor Peter O'Donovan

Please enclosed find our article “Study protocol. Active video games to promote physical activity in children with cancer: a randomized clinical trial with follow-up” with minor changes asked.

Descriptions of the changes made:
- In the method section the participating organizations are named in after title "Research ethics"

Sincerely,

Lotta Kauhanen, the corresponding author